Author's response to reviews

Title: TTP, ET, and ITP sequential occurrence in a single patient: a case report and review of the literature

Authors:

Mirna H Farhat (bellanotchi@hotmail.com)
Philip Kuriakose (pkuriak1@hfhs.org)
Michael Jawad (Mohamadm@umd.umich.edu)
Amr Hanbali (ahanbal1@hfhs.org)

Version: 4 Date: 11 January 2012

Author's response to reviews:

January 11th, 2012
Dear Editor,
Please find enclosed the manuscript entitled below

Title: TTP, ET, and ITP sequential occurrence in a single patient: a case report and review of the literature

Authors: Farhat MH, Kuriakose P, Jawad M, Hanbali A
Changes that were made based on reviewer’s comments were highlighted in red and are as follows:
1-JAK 2 mutation status was added
2-The importance of reviewing peripheral smears in thrombocytopenia evaluation was added to the summary.
3- Consent, competing interest and Author’s contribution sections were moved to before the reference list.
4- Figure was removed from the manuscript body.

Sincerely yours,

Mirna Farhat, MD
Henry Ford Hospital
Detroit, Michigan, 48202